Author's response to reviews

Title: Evolution in functionality of a metastatic pancreatic neuroendocrine tumour (pNET) causing Cushing's syndrome: treatment response with chemotherapy.

Authors:

Surya Panicker Rajeev (panickersurya@yahoo.co.uk)
Steffan McDougall (steffanmcdougall@hotmail.co.uk)
Monica Terlizzo (Monica.Terlizzo@aintree.nhs.uk)
Daniel Palmer (Daniel.Palmer@liverpool.ac.uk)
Christina Daousi (cdaousi@liverpool.ac.uk)
Daniel Cuthbertson (Daniel.Cuthbertson@liverpool.ac.uk)

Version: 2 Date: 7 July 2014

Author's response to reviews:

Dear Sir/Madam,

Many thanks for contacting us about revisions for our manuscript.

We have amended the abstract session which is now divided into background, case presentation and conclusions as advised.

We have also added a background section to the manuscript.

Authors contribution section has also been revised as per instructions. We hope you would kindly consider our manuscript for publication in your journal.

Yours sincerely,

Dr Surya Rajeev, MBBS, MRCP